Author’s response to reviews

Title: Effect of anthelminthic treatment on helminth infection and related anaemia among schoolage children in northwestern Ethiopia

Authors:

Abraham Degarege (abrahamdegarege@yahoo.com)

Yonas Yimam (yonasyimamayene@gmail.com)

Berhanu Erko (berhanue@yahoo.com)

Version: 2 Date: 17 Oct 2016

Author’s response to reviews:

October 16, 2016

Dear Editors of BMC Infectious Disease

Thank you for the comments you have made towards the improvement of this manuscript. We have taken your comments and revised the manuscript accordingly. As you can see below we have provided all sections listed under the heading Declarations in the journals Submission guidelines

Yours sincerely

Abraham

Declarations

Ethics approval and consent to participate

The study obtained ethical approval from the Institutional Review Board of ALIPB. Permission to conduct the study was also obtained from the school directors. Only children who assented and whose parents or guardians orally consented involved in the study.
Consent for publication
Not applicable

Availability of data and materials
The data will not be made available in order to protect the participants’ identity

Competing interest
The authors have declared that no conflict of interests exists.

Funding
This study received financial support from the School of graduate studies and ALIPB, Addis Ababa University.

Authors’ contribution
YY, AD and BE conceived the idea and designed the study; YY and AD collected the data; AD analyzed the data and drafted the manuscript. All authors commented the paper and approved the final manuscript.

Acknowledgements
We would like to thank the laboratory technicians of ALIPB and Jiga Health Center who helped with the fieldwork and sample examination in laboratory. We also acknowledge all the staffs of Tikur Wuha Elementary School community for their assistant during sample collection. We are grateful to school children who participated in this study.